Author’s response to reviews

Title: Prevalence and determinants of chronic kidney disease in rural and urban Cameroonians: A cross-sectional study

Authors:

Francois F Kaze (francoiskaze@yahoo.fr)
Diane T Meto (diane.meto@yahoo.fr)
Marie-Patrice Halle (patricehalle@yahoo.fr)
Jeanne Ngogang (jngogang@yahoo.fr)
Andre-Pascal Kengne (apkengne@yahoo.com)

Version: 4 Date: 16 June 2015

Author’s response to reviews:

MS: 3748860901650201
Prevalence and determinants of chronic kidney disease in rural and urban Cameroonians: A cross-sectional study
Response to reviewer’s comments
Dear Editor,

We are once more grateful to the reviewers for their time and important comments on our manuscript. We also appreciate the invitation and the advice from the editorial office to submit a revised version of our paper.

In line with the editor’s advice, the manuscript has been once more edited by a native-English speaker with scientific expertise.

We look forward to the outcome of your assessment.

For the co-authors,

Dr Kaze